Accelerator Status Report

Run Schedule: Sept. 27, 2010 to December 23, 2010
Nov. 10, 2010
Injector    Energy 61.76 MeV
· Need to set up Wien angle for Hall C.
· (11/4) Beam Studies: injector Wien setup.
· (11/1) Wien was changed from 29 to 5 degrees at Hall A’s request.

· The Gun ramping time was reduced by ~50%. Now it takes about 10 minutes to turn on and ramp up the Gun. 

Accelerator   Linac Energy 549.00 MeV
· Overall the accelerator has been running well.

· (11/10) The north linac 70 MHz MO amplifier faulted, which tripped the NL RF. Ops 

      went out  and reset it and got the RF back.

· (11/10) RF 2L07 misbehaved a few times (~4 hrs). Cavity #7 phase shifted so much that 

            beam would not transport through Arc2. The zone is bypassed for now.
· RF 2L20 tripped off twice and needed to reboot iocsl5 to get back.
· (11/6) PSS BCM dropped the BSY to power due to a +5 power supply (2.35 hrs).
· (11/4) FCup#1 had a waterflow fault. The interlock was jumpered and later removed. 

      Some parts of the system was cleaned up and it works fine now.
· (11/4) LAM3C would not turn off. It was OPS-PRed.
· Optics is good and stable.
· (11/3) Water flow interlock bypassed for FC#1. Chime added to drop mode to beam sync.

· (11/2) Path length drifted and it was corrected.
· (11/2) After running Krest eight cavity phases were adjusted 10 degrees. When manually

      checked them we found one phase adjustment was wrong. 
· RF 2L15 had HV problems a few times. Two of the cavities were bypassed.
· (10/29) ARC6A tripped on DC overload (0.72 hrs.).

· (10/28) MYR9R03 had communication fault and EES fixed it (1.13 hrs.). 

· (10/27) XSEP2T tripped off (0.85 hrs.).
Hall A---Pass 4, DVCS
· The hall will switch to pass 5 sometime this coming week.

· (11/5) Hall A line vacuum went bad while not taking beam. It might be caused by Hall C target cooldown. 
· Hall A beam was fine. 
· FFB feed forward button is on the FFB control screen now. 
· (10/27) Hall A did Moller runs. We did not back out of the Compton.
Hall B ---Pass 5, Primex II
· The hall is happy with its beam. Primex II will be finished on Nov. 11.

· Hall B beam was fine.
Hall C--- Pass 1, QWeak 
· IHA3H09B does not work. EEs working on it.
· IPM3H09B does not work well. Will be replaced.
· (11/7) One of the dump FSD nodes has been masked. 
· Hall C dump signal readbacks are not stable. The experts are aware of it.

· Hall C beamline exit windown has temperature fluctuations  with high current. 

· (11/6) Hall C harp 3H07A found stuck in the beamline.

· (11/5) The hall started to take beam after finishing its target repair.

